Author’s response to reviews

Title: Patients with Pathogenic Variants for Breast Cancer Other Than BRCA1 and BRCA2: Qualitative Interviews about Health Care Experiences

Authors:

Kristin Clift (clift.kristin@mayo.edu)
Sarah Macklin (macklin.sarah@mayo.edu)
Stephanie Hines (hines.stephanie@mayo.edu)

Version: 1 Date: 06 Nov 2019

Author’s response to reviews:

We thank the reviewer for their time and thoughtful comments to improve the manuscript.

1. We added the total number to the abstract

2. We removed all mentions of the word “mutations” and replaced it with “pathogenic variants”

3. We added a list of the other genes in the abstract.

4. We removed the word “mutations” and replaced it with “pathogenic variants”

5. We added the rest of the genes mentioned to the methods and elsewhere in the manuscript and described that these genes are from hereditary breast cancer panels and hereditary breast and gynecologic cancer gene panels.